Author’s response to reviews

Title: Efficient strategy for the molecular diagnosis of intractable early-onset epilepsy using targeted gene sequencing

Authors:

John Hoon Rim (johnhoon1@naver.com)
Se Hee Kim (SEHEEKIM@yuhs.ac)
In Sik Hwang (ISHWANG@yuhs.ac)
Soon Sung Kwon (TNSTNSDLEK@yuhs.ac)
Jieun Kim (jkim1220@gmail.com)
Hyun Woo Kim (PINEBLUE77@yuhs.ac)
Min Jung Cho (MJCHO84@yuhs.ac)
Ara Ko (CHEREARA@yuhs.ac)
Song Ee Youn (YOUNSE71@yuhs.ac)
Jihun Kim (JIHUN264@yuhs.ac)
Young Mock Lee (YMLEEMD@yuhs.ac)
Hee Jung Chung (AGATHAC@yuhs.ac)
Joon Soo Lee (JOONSL96@yuhs.ac)
Heung Dong Kim (HDKIMMD@yuhs.ac)
Jong Rak Choi (CJR0606@yuhs.ac)
Seung-Tae Lee (LEE.ST@yuhs.ac)
Hoon-Chul Kang (hipo0207@yuhs.ac)

Version: 2 Date: 01 Jan 2018

Author’s response to reviews:
Thank you very much for your helpful comments and corrections. We have made revisions according to your comments and suggestions.